Author's response to reviews

Title: The effects of Sutaehwan-Gami on menopausal symptoms induced by ovariectomy in rats

Authors:

Seung-bok Lee (jepark1999@hanmail.net)
Songhee Jeon (jsong0304@dongguk.edu)
Sok Cheon Pak (spak@csu.edu.au)
Dong-II Kim (obgykdi@hanmail.net)
Min Sun Choi (gene11@hanmail.net)

Version: 5 Date: 22 October 2012

Author's response to reviews: see over
Title: The effects of Sutaehwan-Gami on menopausal symptoms induced by ovariectomy in rats

Version: 3

Date: 22 October 2012

Reviewer: Gail B Mahady

Reviewer’s report:
Compulsory rewrite and extensive English Editing

Level of interest: An article whose findings are important to those with closely related research interests

Quality of written English: Not suitable for publication unless extensively edited

Statistical review: No, the manuscript does not need to be seen by a statistician.

➤ As you recommended above, we extensively edited the manuscript by a native-English speaker with scientific expertise.
➤ Abstract format was changed same as the journal format.